Author’s response to reviews

Title: Rapid Diagnosis of Propionibacterium acnes Infection in Patient with Hyperpyrexia after Hematopoietic Stem Cell Transplantation by Next-Generation Sequencing: A Case Report

Authors:
Mingzhi Ye (yemzchina@163.com)
Wei Wei (weiwei3@genomics.cn)
Zhikai Yang (568329311@qq.com)
Yingzhen Li (liyingzhen@genomics.cn)
Shaomin Cheng (chengshaomin@genomics.cn)
Kang Wang (wangkang1@genomics.cn)
Tianliangwen Zhou (408915649@qq.com)
Jingmeng Sun (840796596@qq.com)
Sha Liu (972930260@qq.com)
Na Ni (nina@genomics.cn)
Hui Jiang (jianghui@genomics.cn)
Hua Jiang (weimeitiantang1213@163.com)

Version: 3 Date: 27 Oct 2015

Author’s response to reviews:

1. Please clarify in the manuscript who granted permission to publish the manuscript as currently it reads that the patient did. Did the patient's parents provide consent to publish the manuscript?

   A1: Thank you for pointing this out. We had clarified this section in Consent. And we had got the written informed consent for publication of this case report from the patient's mother (in the attach files).

2. Specifically mention the to the case and the analysis of each author by their unique initials in the Authors' Contributions section.
A2: Thank you for pointing this out. We have modified it in Authors’ contributions.

3. List the figure legends after the References.

A3: As suggested, we have modified this section.

4. Upload an English version of the CARE checklist.

A4: Thank you. We have uploaded it in the attach files.